Author’s response to reviews

Title: Two monogenic disorders masquerading as one: Severe congenital neutropenia with monocytosis and non-syndromic sensorineural hearing loss

Authors:
Parvathy Venugopal (parvathy.venugopal@sa.gov.au)
Lucia Gagliardi (Lucia.Gagliardi@sa.gov.au)
Cecily Forsyth (cecily.forsyth@healthemail.com.au)
Jinghua Feng (jinghua.feng@unisa.edu.au)
Kerry Phillips (Kerry.Phillips@sa.gov.au)
Milena Babic (milena.babic@sa.gov.au)
Nicola Poplawski (Nicola.Poplawski@sa.gov.au)
Hugh Rienhoff (hrienhoff@datahooks.com)
Andreas Schreiber (andreas.schreiber@adelaide.edu.au)
Christopher Hahn (chris.hahn@sa.gov.au)
Anna Brown (anna.brown@sa.gov.au)
Hamish Scott (hamish.scott@sa.gov.au)

Version: 3 Date: 06 Feb 2020

Author’s response to reviews:

Dear Editor,

We have included the web links and accession numbers for the data in the manuscript. The study has now been released by EGA as per the requirements of the journal.

We have also uploaded final clean versions of the manuscript, figure, table and supplementary material.

We appreciate the journal’s efforts in handling this manuscript.

Kind regards,

Prof. Hamish Scott